\begin{thebibliography}{10}

\bibitem{RTCorba:02}
{Object Management Group}, {\em {Real-time CORBA Specification}}, {OMG Document
  formal/02-08-02}~ed., Aug. 2002.

\bibitem{RTSJ:00}
G.~Bollella, J.~Gosling, B.~Brosgol, P.~Dibble, S.~Furr, D.~Hardin, and
  M.~Turnbull, {\em The Real-time Specification for Java}.
\newblock Addison-Wesley, 2000.

\bibitem{Sharp:03a}
{\em Evaluating Real-time Java for Mission-Critical Large-Scale Embedded
  Systems}, (Washington, DC), IEEE Computer Society, May 2003.

\bibitem{Subramonian:04c}
V.~Subramonian, L.-J. Shen, C.~Gill, and N.~Wang, ``{The Design and Performance
  of Dynamic and Static Configuration Mechanisms in Component Middleware for
  Distributed Real-time and Embedded Systems},'' in {\em The 25th IEEE
  Real-time Systems Symposium (RTSS)}, (Lisbon, Portugal), Dec. 2004.

\bibitem{LWCCM}
{Object Management Group}, {\em {Lightweight CCM RFP}},
  {realtime/02-11-27}~ed., Nov. 2002.

\bibitem{Schmidt:03m}
N.~Wang, D.~C. Schmidt, A.~Gokhale, C.~Rodrigues, B.~Natarajan, J.~P. Loyall,
  R.~E. Schantz, and C.~D. Gill, ``{QoS-enabled Middleware},'' in {\em
  Middleware for Communications} (Q.~Mahmoud, ed.), pp.~131--162, New York:
  Wiley and Sons, 2003.

\bibitem{Sharp:98}
D.~C. Sharp, ``{Reducing Avionics Software Cost Through Component Based Product
  Line Development},'' in {\em Proceedings of the 10th Annual Software
  Technology Conference}, Apr. 1998.

\bibitem{Ritter:03}
T.~Ritter, M.~Born, T.~Untersch\"{u}tz, and T.~Weis, ``{A QoS Metamodel and its
  Realization in a CORBA Component Infrastructure},'' in {\em {Proceedings of
  the $36^{th}$ Hawaii International Conference on System Sciences, Software
  Technology Track, Distributed Object and Component-based Software Systems
  Minitrack, HICSS 2003}}, (Honolulu, HW), HICSS, Jan. 2003.

\bibitem{Sharp:99}
D.~C. Sharp, ``{Avionics Product Line Software Architecture Flow Policies},''
  in {\em Proceedings of the 18th IEEE/AIAA Digital Avionics Systems Conference
  (DASC)}, Oct. 1999.

\bibitem{Sharp:03}
D.~C. Sharp and W.~C. Roll, ``{Model-Based Integration of Reusable
  Component-Based Avionics System},'' in {\em Proc. of the Workshop on
  Model-Driven Embedded Systems in RTAS 2003}, May 2003.

\bibitem{Greenfield:04}
J.~Greenfield, K.~Short, S.~Cook, and S.~Kent, {\em Software Factories:
  Assembling Applications with Patterns, Models, Frameworks, and Tools}.
\newblock New York: John Wiley \& Sons, 2004.

\bibitem{Janos:03}
G.~Karsai, J.~Sztipanovits, A.~Ledeczi, and T.~Bapty, ``Model-integrated
  development of embedded software,'' {\em Proceedings of the IEEE}, vol.~91,
  pp.~145--164, Jan. 2003.

\bibitem{Schmidt:04a}
A.~Gokhale, K.~Balasubramanian, J.~Balasubramanian, A.~S. Krishna, G.~T.
  Edwards, G.~Deng, E.~Turkay, J.~Parsons, and D.~C. Schmidt, ``{Model Driven
  Middleware: A New Paradigm for Deploying and Provisioning Distributed
  Real-time and Embedded Applications},'' {\em The Journal of Science of
  Computer Programming: Special Issue on Model Driven Architecture}, 2005 (to
  appear).

\bibitem{Schmidt:05y}
G.~Deng, J.~Balasubramanian, W.~Otte, D.~C. Schmidt, and A.~Gokhale, ``{DAnCE:
  A QoS-enabled Component Deployment and Conguration Engine},'' in {\em
  Proceedings of the 3rd Working Conference on Component Deployment},
  (Grenoble, France), Nov. 2005.

\bibitem{DandC:03}
{Object Management Group}, {\em {Deployment and Configuration Adopted
  Submission}}, {OMG Document ptc/03-07-08}~ed., July 2003.

\bibitem{Schmidt:04z}
K.~Balasubramanian, J.~Balasubramanian, J.~Parsons, A.~Gokhale, and D.~C.
  Schmidt, ``{A Platform-Independent Component Modeling Language for
  Distributed Real-time and Embedded Systems},'' in {\em Proceedings of the
  11th Real-time Technology and Application Symposium (RTAS '05)}, (San
  Francisco, CA), pp.~190--199, IEEE, Mar. 2005.

\bibitem{GME:01}
A.~Ledeczi, A.~Bakay, M.~Maroti, P.~Volgysei, G.~Nordstrom, J.~Sprinkle, and
  G.~Karsai, ``{Composing Domain-Specific Design Environments},'' {\em IEEE
  Computer}, pp.~44--51, November 2001.

\bibitem{Schmidt:04e}
G.~Edwards, D.~C. Schmidt, A.~Gokhale, and B.~Natarajan, ``{Integrating
  Publisher/Subscriber Services in Component Middleware for Distributed
  Real-time and Embedded Systems},'' in {\em Proceedings of the 42nd Annual
  Southeast Conference}, (Huntsville, AL), ACM, Apr. 2004.

\bibitem{Lee:01}
J.~Liu, X.~Liu, and E.~A. Lee, ``{Modeling Distributed Hybrid Systems in
  Ptolemy II},'' in {\em Proceedings of the American Control Conference}, June
  2001.

\bibitem{MDA:01}
{Object Management Group}, {\em {Model Driven Architecture (MDA)}}, {OMG
  Document ormsc/2001-07-01}~ed., July 2001.

\bibitem{KC:04}
K.~Carter, ``{Kennedy Carter iUML 2.2}.'' \url{www.kc.com}, 2004.

\bibitem{Karsai:02b}
G.~Karsai, S.~Neema, B.~Abbott, and D.~Sharp, ``{A Modeling Language and Its
  Supporting Tools for Avionics Systems},'' in {\em Proceedings of 21st Digital
  Avionics Systems Conf.}, Aug. 2002.

\bibitem{Wendy:03}
W.~Roll, ``{Towards Model-Based and CCM-Based Applications for Real-time
  Systems},'' in {\em {Proceedings of the International Symposium on
  Object-Oriented Real-time Distributed Computing (ISORC)}}, IEEE/IFIP, May
  2003.

\bibitem{Hatcliff:03}
J.~Hatcliff, W.~Deng, M.~Dwyer, G.~Jung, and V.~Prasad, ``{Cadena: An
  Integrated Development, Analysis, and Verification Environment for
  Component-based Systems},'' in {\em Proceedings of the 25th International
  Conference on Software Engineering}, (Portland, OR), May 2003.

\bibitem{GReAT:03}
{Karsai G. and Agrawal A. and Shi F. and Sprinkle J.}, ``{On the use of Graph
  Transformations in the Formal Specification of Computer-Based Systems},'' in
  {\em Proceedings of IEEE TC-ECBS and IFIP10.1 Joint Workshop on Formal
  Specifications of Computer-Based Systems}, (Huntsville, AL), IEEE, Apr. 2003.

\end{thebibliography}

% $Id: relwork.tex,v 1.7 2006/01/04 19:35:52 dengg Exp $

\section{Related Work}
\label{sec:relwork}

Our work on MDD extends earlier work on Model-Integrated Computing
(MIC).  Examples of MIC technology used today include
GME~\cite{GME:01} and Ptolemy~\cite{Lee:01} (used primarily in
real-time and embedded domains) and Model Driven Architecture
(MDA)~\cite{MDA:01} based on UML and XML. Kennedy Carter's iUML
Product Suite~\cite{KC:04} supports the Executable UML process from
textual requirements management through modeling to complete target
code generation. The Rhapsody System Designer tool by I-Logix
(www.ilogix.com) is based on OMG's MDA and UML 2.0 specification,
and provides complete application generation from UML models, and
the generated application could be in multiple programming languages
including C/C++, Java or Ada, and multiple middleware platforms
including CORBA and COM. Both tools have proved many successes in
developing DRE systems. Unlike our approach, however, these MDD
tools are based on OMG's UML specification which is still in the
low-level solution space. In contrast, our work combines GME
metamodeling mechanisms and UML capability to model and synthesize
component middleware for configuring and deploying DRE systems,
which better unifies the problem space and solution space.

The {\em Embedded Systems Modeling Language} (ESML)~\cite{Karsai:02b}
was developed by ISIS at Vanderbilt University to provide a visual
metamodeling language that captures multiple views of embedded
systems, allowing a diagrammatic specification of complex models. The
modeling building blocks include software components, component
interactions, hardware configurations, and scheduling policies.  The
user-created models can be fed to analysis tools (such as AIRES, VEST,
and Cadena) to perform schedulability and event analysis. Using these
analyses, design decisions (such as component allocations to the
target execution platform) can be performed. Unlike PICML, ESML is
platform-specific since it is customized for the Boeing Bold Stroke
PRiSm QoS-enabled component model~\cite{Sharp:03,Wendy:03}.  ESML also
does not support nested assemblies and the allocation of components
are tied to processor boards, which is a proprietary feature of the
Bold Stroke component model.

Cadena~\cite{Hatcliff:03} is a MDD tool developed within Eclipse
Modeling Framework (EMF) for building component-based DRE systems,
with the goal of applying static analysis, model checking, and
lightweight formal methods to enhance these systems. Unlike our work
on CoSMIC, however, Cadena does not support activities such as
component packaging, generating deployment plan descriptors, and
hierarchical modeling of component assembly, thus it introduces
additional burden to DRE application developers to accomplish such
tasks. In our work, these aspects could be captured through PICML MDD
tool and then all the deployment and configuration work can be
automated using DAnCE. We are collaborating with the Cadena
researchers to create an integrated suite of MDD
tools~cite{Schmidt:04v}.

%%% Local Variables:
%%% mode: latex
%%% TeX-master: "isorc06-its"
%%% End:
